Article details: 2015-0081				
	Childhood obesity prevention interventions in primary care: a qualitative study of			
Title	perspectives of primary care clinicians and parents of 2-5 year old children			
Authors	Nicole Bourgeois MSc RD, Paula Brauer, PhD, RD, Janis Randall Simpson, PhD, RD, Susie Kim, MD, CCFP, Jess Haines, PhD, RD			
Reviewer 1	Hasanain Ghazi			
Institution	Kuala Lumpur, Malaysia			
Reviewer 1				
	proof page 14 (lines 39-43) to better highlight this. It now reads: "Our sample size for parent participants was relatively small, and parents were not diverse with respect to			
	age, gender or ethnicity despite an extended recruitment phase, which limits the generalizability of our findings."			
Reviewer 2	Jabir Jassam			
Institution	Merrickville District Community Health Centre, Family Medicine, Merrickville, ON			
General comments (author response in	What program you can apply in this age group? I might be more interested if the age is between 5-10 RESPONSE: Thank you for this question. Our team acknowledges that in order to be			

bo	l	d)	۱
$\omega$	ľ	u,	

effective on a population level, childhood obesity prevention interventions are needed for all age groups. In the current study, our target population focused on families of 2-5 year old children, given that obesity is observed this early, and that there are few preventative interventions specifically designed for this age group. We explored the perspectives of clinicians and parents through the lens of an existing obesity prevention intervention, which was designed specifically for families with children 2-5 years of age; behaviours addressed within the intervention are designed to be developmentally appropriate for this age group.

2. The authors are not MDs except one which makes their experience in the primary care setting is limited

**RESPONSE:** Thank you for this comment. While only one of the authors is a primary care MD, the first author (NB) is a primary care registered dietitian, and second author's primary research area involves nutrition services in primary care settings (PB). In order to enhance reflexivity with respect to the research team, additional information on NB's role has been added under "Data Collection & Analytic Approach" (pdf proof pg 7, lines 41-51; pg 8, lines 9-13).

3. The number of primary MDs who participated is very low

**RESPONSE:** Thank you for this comment. In our study, we had 6 MD participants in our focus groups; as a group, they were second most common participants. Our setting and sites represent one model of primary care which is team-based and interprofessional. In these settings, it is common for RNs as well as NPs to be highly involved in the provision of well-child care, and obesity prevention efforts. As such, adequate representation from RNs and NPs is essential to understand the context and setting of our results. We agree that it is important to note the limitations of conceptual generalizability of our results, in that they cannot be applied to understand other models of primary care (i.e., solo family MDs). We have added some content to our "Limitations" section to identify the limits to generalizability of our results (pdf proof pg 14, lines 33-39). It now reads: "While we engaged a variety of primary care clinicians from three different settings (urban academic, urban, and rural), we only recruited from the FHT model, which does not reflect the diversity of practices in Canada, and limits the generalizability of our findings.